Author's response to reviews

Title: Trends in antibiotic use among outpatients in New Delhi, India

Authors:

Anita Kotwani (anitakotwani@gmail.com)
Kathleen Holloway (hollowayk@searo.who.int)

Version: 4 Date: 7 April 2011

Author's response to reviews: see over
April 7, 2011

Dear Editor,

I am submitting the revised manuscript R3 (MS: 1694694726487911), entitled, “Trends in antibiotic use among outpatients in New Delhi, India”.

We have revised the manuscript on the lines suggested by the reviewer for minor language issues and the additional editorial requirement is also incorporated. The changes done are as track changes in the revised MS.

Hope you will find the MS in order.

A point-by-point response to the concerns of the reviewer and editorial office is mentioned below.

Many Thanks.

Best Regards,

Anita Kotwani

Authors’ response to reviewer

Title: Trends in antibiotic use among outpatients in New Delhi, India
Authors: Anita Kotwani and Kathleen Holloway

Reply to comments of Reviewer 1

Thanks for appreciating the hard work to improve the paper.

Some minor language issues

The minor language issues mentioned in the abstract, page 6, Page 8, page 9, and page 10 are corrected.

Editorial Office

We have copy edited the paper.